Author's response to reviews

Title: Bilateral spontaneous non-traumatic rupture of the Achilles tendon - a case report

Authors:

Zubair Khanzada (zubairshabbir@hotmail.com)
Ulfin Rethnam (ulfir@yahoo.com)
David Widdowson (David.Widdowson@wales.nhs.uk)
Ahmad Mirza (Ahmad.Mirza@wales.nhs.uk)

Version: 4 Date: 2 July 2010

Author's response to reviews:

Please find the list of changes made as below.

Major revisions:
Author list: Has been amended as suggested.
Abstract: The manuscript abstract and the submission abstract have been made identical.
Ethnicity: The term 'white' has been taken out.
Authors' contributions: Author's contributions have been included.

Minor revisions:
Title page: Superscript numbers used to link authors to the affiliations. Full stop removed from the title.
Structure: Figure legends section moved to after the references.
References: The term 'et al' replaced with full reference.
Figures: Title removed. The legend and title moved to after the reference list